



Article Exploring the Influence of Environmental Values on Green Consumption Behavior of Apparel: A Chain Multiple Mediation Model among Chinese Generation Z

Jianfang Liang ^{1,*}, Jingjun Li² and Qinyuan Lei¹

- ¹ School of Fashion and Art Design, Xi'an Polytechnic University, Xi'an 710048, China
- ² Research Group MOBI, Department of Business Technology and Operations, Vrije Universiteit Brussel, Pleinlaan 2, 1050 Brussel, Belgium
- Correspondence: liangjianfang69@163.com

Abstract: There is a problem among Generation Z regarding the insufficient perception of green apparel consumption and the "perception-action paradox", which presents a great challenge to China's future sustainable development. To address this problem, we constructed a chain multiple mediation research framework that explored the transmission paths between the environmental value and green consumption behavior of apparel, as well as the associated influence mechanisms; this was performed by integrating environmental responsibility and green consumption intention. Data for this study were collected through a multistage sampling survey of 657 Chinese Gen Z members born between 1995 and 2002. Our results reveal that all three types (egoism, altruism, and biospheric values) of environmental values had different direct and indirect effects on the green apparel consumption behavior for Gen Z members, but the indirect effects of each aspect significantly outweighed their direct effects. In terms of the direct effects, egoistic values had no significant direct negative effect on green apparelconsumption behavior, whereas the significant positive direct effect of the biospheric values was greater than that of altruistic values. The greatest mediating effect between the environmental values and green apparelconsumption behaviorwas green consumption intention, followed by environmental responsibility, and ending with the chain mediation effect of environmental responsibility and green consumption intention. These findings suggest that it is imperative to stress the cultivation of green consumption intentions, environmental responsibility, and environmental values and bridge the seamless link among these variables for the promotion of green apparel consumption practices in Generation Z. This is the first study that explicitly identifies the significant chain mediating effect of environmental responsibility and green consumption intention between the environmental values and green apparel consumption behavior. Our findings broaden the theoretical research perspective of green apparel consumption behavior and provide a reference for the guiding of green consumption practices and policy formulation for the global population of Generation Z.

Keywords: environmental value (ENV); green consumption behavior of apparel (GCBA); environmental responsibility (ER); green consumption intention (GCI); chain multiple mediation

Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). 1. Introduction

As the environment continues to deteriorate, environmental problems have attracted public attention worldwide, and green consumption has become the strongest voice of the times. More and more consumers are becoming more conscious about environment-related issues [1] and take sustainable consumption as their responsibility and behavioral intentions. A survey by Nielsen covering more than 30,000 consumers from 60 countries in 2015 demonstrated that nearly three-quarters of individuals aged 19–34 years (as of 2015) would like to pay more for sustainable products [2]. In China, over 90% of consumers are aware of sustainable consumption, and 70% of consumers are very conscientious



Citation: Liang, J.; Li, J.; Lei, Q. Exploring the Influence of Environmental Values on Green Consumption Behavior of Apparel: A Chain Multiple Mediation Model among Chinese Generation Z. *Sustainability* **2022**, *14*, 12850. https:// doi.org/10.3390/su141912850

Academic Editors: Ming-Lang Tseng, Yeneneh Tamirat Negash and Tat-Dat Bui

Received: 14 September 2022 Accepted: 5 October 2022 Published: 9 October 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.

of it [3]. However, growing public concerns about environmental issues have not led to a parallel increase in the consumption of green products over the past few years [4]. The global market share of green products was less than 4% in 2013 [4], and sustainable products only contributed 20% to the total consumption of China in 2021 [3]. Taking the apparel industry as an example, the prevalence of consumerism has stimulated the idea of buying new apparel regularly [5] and led to the trend of "fast fashion". "Fast fashion" is characterized by a shortened fashion cycle, presenting cheap manufacturing, more frequent consumption, and a short-lived use and disposal of apparel items [6]. This trend has promoted the formation of rapid response systems [7] and flexible supply chains in the apparel industry [8]. However, it has also resulted in a considerable waste of resources and serious environmental pollution problems in the apparel industry [9,10], accounting for approximately 8–10% of global carbon emissions [11].

As the largest developing economy, China has been playing a dominant role in the global apparel industry, and the Chinese apparel market has been projected to reach a value of USD 615 billion by 2025 [12]. In the context of "carbon peaking and carbon neutrality", many Chinese consumers have expressed their intentions to reduce clothing consumption and purchase green clothing as much as possible [8]. However, the reality is that they still enjoy over-consumption and are facing the dilemma of green clothing consumption [13]. This is particularly evident among members of Generation Z (abbreviated to Gen Z) [14], who are a potent force for eco-friendly practices in the apparel industry [8]. Undoubtedly, the apparel industry in China is facing a huge challenge in terms of commitment and action towards sustainability.

Gen Z, whose members are born between the years of 1995 and 2012 [15], is a group of digital natives who are extensively engaged with technology [16]. They are full of confidence, loving self-expression, adventure, and more eager to achieve their values [17]; they are considered to be the most powerful driver of consumption [18], contributing approximately USD 323 billion alone to the US, and comprising 40% of US consumers in 2021 [8]. As they are well-educated and actively utilize social media [19], they are good at accepting new concepts (e.g., collaborative clothing consumption [20]). Compared with other generations, Gen Z is more knowledgeable about sustainable living and gives more priority to environmentally friendly (often referred to as green) products [21]. Hence, Gen Z is considered to be the most motivated potential and influential generation among all generations in terms of sustainable consumption, representing about 280 million people in China [17]. Meanwhile, Gen Z, characterized as fashion trendsetters, tends to use clothing as an impression management tool; the members prefer to reflect their social status, prestige, and success through material ownership and are inclined to instant gratification consumption [13] to satisfy their hedonic and fashion needs [22]. They are more likely to be involved in compulsive buying and the so-called "mianzi" consumption of luxury goods [23] rather than adopt an environmentally friendly lifestyle [24,25]. Hence, they face difficulties integrating green behavior into their daily lives [26], showing a "perceptionaction paradox". On the one hand, they offer a strong vital concern for environmental issues and seem to hold well-grounded environmental values. On the other hand, they are deeply involved in over-consumption [27,28] and non-environmentally friendly consumption [1]. This implies that the positive environmental values of Gen Z do not inspire them to develop a higher sense of environmental responsibility and contribute to green consumption intentions and behaviors. Then, what causes the gap between the environmental values and green consumption behaviors in Gen Z? Are environmental responsibility and green consumption intentions important transmitting variables between environmental values and green consumption behavior? What is the influence mechanism between them?

Regarding the green consumption of Gen Z, researchers in many countries have made significant contributions. The existing studies available mainly include the following research focuses. One focus has been the current status of green consumption in specific categories of products, such as the green consumption of food and household appliances. Another focus has been the influencing factors of green consumption, such as the consumers' personality, knowledge, socio-demographic characteristics, consumers' pro-environmental attitudes, etc. Previous research has examined the direct influence of consumers' environmental values on green consumption, but the "perception–action paradox" is still widely found among Gen Z worldwide, suggesting that there are likely other transmission paths between them; could it be the influence of environmental responsibility and green consumption intention? In addition, the green consumption behavior of clothing is more complex than other types of environmentally friendly products [1], involving the influence of multidimensional variables such as personal, psychological, social, cultural, and apparel-related attributes, such as style, trend, and fit. To the best of our knowledge, the relationship between environmental values and green consumption behavior of apparel, as well as the influence mechanism of the presence of environmental responsibility and green consumption intention have rarely been addressed in the literature. This study takes Chinese Gen Z as its research subject and explores the internal psychological mechanisms present by testing the respective and chain-mediating roles of the studied variables, aiming

The main research questions of this study were as follows:

sustainable consumption around the world.

(1) How do environmental values influence the green clothing consumption behavior of Gen Z?

to provide a corresponding decision basis and policy suggestions for the promotion of

- (2) How do environmental responsibility and green consumption intention impact green clothing consumption behavior?
- (3) Does the relationship between the environmental values and green consumption behavior of apparel depend on the presence of environmental responsibility and green consumption intention? If so, in what way?

The remainder of the paper is structured as follows: Section 2 provides a comprehensive literature review and the hypothesis development; Section 3 introduces the research methodology; Section 4 presents the results of the empirical investigation in detail; and Section 5 provides a discussion of all findings. Section 6 puts forward the study's theoretical contributions and managerial insights, which is then closed with the conclusions and study limitations being listed in Section 7.

2. Literature Review and Hypotheses Development

2.1. Green Consumption Behavior of Apparel (GCBA)

Green consumption, often related to environmentally sustainable consumption [29], is an effective measurement that reduces the negative impacts of consumption on the environment and eco-systems [30]. It details the consumers' responsibility of adopting environmentally friendly behaviors to attain compatibility with safeguarding the environment between the present and future generations [31]. In this regard, green consumption behavior of apparel, characterized by resource conservation and environmental protection, is mainly manifested by a clothing consumption reduction, the usage and recycling of second-hand clothing, and the purchase of green apparel products; thus, it reduces the negative impacts of apparel consumption to the environment during the manufacturing, usage, and recycling processes [32,33]. Among them, the purchase of green apparel is considered the best approach over the others [34]. This is because green apparel is characterized by its use of organically grown fibers, recycled materials, having no dying process, having environmentally friendly labeling or packaging techniques, and having a long life [32]; these aspects are beneficial to the environment and ecology in more than one stage of their respective life cycles [35,36].

Although green clothing reduces their environmental impact and offers other similar benefits such as functional benefits [37], the consumption of green clothing has not met expectations due to the inadequate awareness of consumers [21], general premium of conventional clothing [38], and certain sacrifices in personal interests [39]. Only a small proportion of consumers regard sustainability as a primary criterion when evaluating alternatives in the process of buying apparel [40]. At the same time, most customers are still

influenced by the hedonism of sensory experiences and the beauty of apparel in their consumption [41]. A few studies focusing on green apparel have explored several factors, such as consumer perception [42], values (fashion involvement) [1], budget constraints [33,36], reference groups [25], etc.; the question, however, remains why consumers do not adopt green consumption practices [43]. Therefore, exploring the determinants of the green consumption of apparel for the young generation is an important and urgent task.

2.2. The Influence of Environmental Values (ENV) on Green Consumption Behavior of Apparel

Values, the guiding principles in people's lives [44,45], determine what people want to engage in, what knowledge and perceptions they want to acquire most, how people evaluate various aspects of situations, and how they consider and choose behavioral alternatives, which in turn affects their actual behavior [46]. The existing studies revealed that values had a predominant influence on individual environment-friendly behaviors [47]; however, values differ between different individuals and contexts. This implies that people tend to make choices and act according to the values they consider most important, especially when facing conflicting values. Different people make different choices based on the priority of various values [46].

As a special category of values, environmental values provide individuals with criteria and standards for viewing environmental issues and are regarded as an important cornerstone for interpreting various pro-environmental behaviors in-depth. Currently, the three types of values proposed by Stern et al. [48] are widely recognized: egoistic, altruistic, and biospheric values. These values have been clearly distinguished by De Groot et al. [49,50]. Egoistic values especially emphasize costs and benefits from an individual's environmental actions for personal gain. Only when the perceived benefits exceed the perceived costs will the individuals generate an environmentally friendly intention and vice versa. Altruistic values focus on the outcomes of actions for other people, while biospheric values highlight the perceived costs and benefits of an individual's environmentally friendly behavior for the ecosystem and biosphere as a whole [50]. Several studies have examined environmental values as predictors of green consumption behavior, which revealed that individuals who endorsed altruistic and particularly biospheric values were more likely to engage in pro-environmental behavior compared with those who strongly endorsed egoistic values [50,51]. In the apparel field, Bielawska and Grebosz-Krawczyk found that environmental values had a significantly positive impact on consumers' choice of behavior regarding green clothing products [33]. Hong et al. [52] addressed that altruistic values significantly affected the post-purchase satisfaction of green apparel products and influenced the willingness to repurchase green apparel products. However, it is still unsolved how all three values influence the green consumption of apparel for Chinese Gen Z members, and which is most important for understanding the green consumption of apparel. Based on the discussion above, the following hypotheses were proposed.

Hypothesis 1 (H1). *Environmental values significantly influence green consumption behavior of apparel.*

Hypothesis 1a (H1a). Egoistic values significantly influence green consumption behavior of apparel.

Hypothesis 1b (H1b). Altruistic values significantly influence green consumption behavior of apparel.

Hypothesis 1c (H1c). *Ecological values significantly influence green consumption behavior of apparel.*

2.3. Mediating Effect of Environmental Responsibility (ER)

Regarding environmental responsibility, most of the literature refers to the corporate environmental responsibility [53] rather than consumer environmental responsibility [54], which we shall discuss in this study. Stone et al. [55] defined environmental responsibility (ER) as, "a state in which a person expresses an intention to take action directed toward remediation of environmental problems—acting not as an individual consumer with his or her own economic interests, but through a citizen-consumer concept of societalenvironmental well-being". This implies that environmental responsibility enables people to confront the negative impact of their behavior on the environment and resources, internalize their perception and responsibility toward the environment, and then change their decision-making tendencies and behavioral habits, resulting to an active practice of pro-environmental behavior. The importance of environmental responsibility to green consumption behavior is evident.

Slavoljub et al. have confirmed the positive linear dependence between environmental values and personal environmental responsibility among young people in Serbia [56]. Barr and Gilg [57] found that committed environmentalists developed a high level of concern and value for environmental issues and demonstrated a willingness to accept a personal responsibility and moral obligation for improving the environment. Thus, we proposed the following hypotheses.

Hypothesis 2 (H2). Environmental values exert a significant effect on environmental responsibility.

Hypothesis 2a (H2a). Egoistic values exert a significant effect on environmental responsibility.

Hypothesis 2b (H2b). Altruistic values exert a significant effect on environmental responsibility.

Hypothesis 2c (H2c). Ecological values exert a significant effect on environmental responsibility.

Meanwhile, it has been confirmed that an individual's environmental responsibility is the most fundamental antecedent variable of their pro-environmental behavior [48], which influences all types of environmental behaviors [58,59]. Kaiser and Scheuthle [60] found a positive relationship between consumer environmental responsibility and environmentally friendly behavior in Swiss citizens. A few research studies have demonstrated that environmental responsibility has been considered a mediating variable to explain consumers' pro-environmental behaviors. Shahrin et al. [61] examined the mediating role of environmental responsibility between "compensatory health beliefs and pro-environmental behavior" and "environmental self-identity and pro-environmental behavior" in nutricosmetics consumption.

Although past studies have examined the direct and indirect effects of environmental values on green purchasing behavior, current research is not available on the mediating role of environmental responsibility in the relationship between environmental values and the green consumption behavior of clothing. With the awakening of environmental awareness in recent years, Gen Z has been encouraged to establish better environmental values; they tend to exhibit higher responsibility and commitment in solving environmental problems, in turn enhancing the individual green consumption behavior of clothing. Based on the discussion above, we inferred the following hypotheses.

Hypothesis 3 (H3). *In the presence of environmental responsibility, environmental values have an influence on the green consumption behavior of apparel.*

Hypothesis 3a (H3a). *In the presence of environmental responsibility, egoistic values have an influence on the green consumption behavior of apparel.*

Hypothesis 3b (H3b). *In the presence of environmental responsibility, altruistic values have an influence on the green consumption behavior of apparel.*

Hypothesis 3c (H3c). *In the presence of environmental responsibility, biospheric values have an influence on the green consumption behavior of apparel.*

2.4. The Mediating Effect of Green Consumption Intention (GCI)

Green consumption intention is defined as the willingness of individuals to purchase eco-friendly products (rather than conventional products) and their related services [62]. Many empirical studies [63] and theories (such as TPB and TRA) [64] on green consumption have shown that green consumption intention and green consumption behavior are highly correlated. Therefore, green consumption intention is an essential key indicator in the study of the green consumption behavior of apparel. Green consumption intention is a manifestation of voluntary behavior, and it is influenced by the two important drivers of willingness and intensity of effort to perform [65]. It has been demonstrated that the more consumers are willing to understand eco-friendly products, the higher their intention to use green products [66]. Environmental values are one of the factors influencing green consumption intentions and have a positive effect on these intentions [43,67]. Varshneya et al. [26] reported that green consumption values positively influence the behavioral intention for organic clothing. Thus, we developed the following hypotheses.

Hypothesis 4 (H4). Environmental values significantly influence green consumption intention.

Hypothesis 4a (H4a). Egoistic values significantly influence green consumption intention.

Hypothesis 4b (H4b). Altruistic values significantly influence green consumption intention.

Hypothesis 4c (H4c). Biospheric values significantly influence green consumption intention.

In fact, many studies have confirmed that behavioral intention is an important factor that influences an individual's actual consumption behavior, and it acts in various roles as an antecedent, mediator, and outcome variable. Behavioral intention has been demonstrated as a direct antecedent variable of environmental behavior [68,69] and is widely used in studies of organic food, electric vehicles, energy-efficient appliances, and green clothing [70]. The theory of reasoned action (TRA) demonstrates that behavioral intention is deemed to be an immediate antecedent of behavior [64], and some researches have verified that purchase intention exerts a positive effect on the purchase behavior of organic products [71,72]. Based on the TPB model, green product consumption intention completely mediates between green product purchase behavior and its antecedent predictor variables (i.e., attitudes toward green products, subjective norms, and perceived behavioral control) [13]. The positive effect of green consumption intention on green consumption behavior has also been validated [73,74].

In this study, we deduced that Chinese Gen Z members, who value self-expression, identity, and group affiliation, should demonstrate a positive value orientation toward environmental and ecological issues, show a high intention of green consumption [75], and should be more inclined to implement green clothing consumption behaviors. Therefore, the following hypotheses were developed.

Hypothesis 5 (H5). *In the presence of green consumption intention, environmental values have an influence on the green consumption behavior of apparel.*

Hypothesis 5a (H5a). *In the presence of green consumption intention, egoistic values have an influence on the green consumption behavior of apparel.*

7 of 26

Hypothesis 5b (H5b). *In the presence of green consumption intention, altruistic values have an influence on the green consumption behavior of apparel.*

Hypothesis 5c (H5c). *In the presence of green consumption intention, biospheric values have an influence on the green consumption behavior of apparel.*

2.5. Chain Mediating Effect of Environmental Responsibility and Green Consumption Intention

Environmental responsibility originates from the normative activation model in social psychology and has been widely used in consumer behavior research [56]. As mentioned above, the positive contribution of the consumer's environmental responsibility to green consumption behavior has been proven in other countries [60]. Only a few studies that have been conducted on the relationship between environmental responsibility and green consumption intentions suggest that environmental responsibility is an important factor influencing green consumption intentions. For example, Attaran and Celik [76] revealed that individuals with a high level of environmental responsibility were more likely to exhibit favorable attitudes and purchase intentions toward green buildings in the United States. Yue et al. concluded in their study that consumers' environmental responsibility significantly affects their green consumption intention in the Chinese context [77]. Therefore, the environmental responsibility is not only an important factor influencing green consumer behavior, but also an important factor that drives green consumption intentions. When consumers intentionally try to solve environmental issues, environmental responsibility will become an obligation of that individual, which tends to be an important predictor of green consumption behavior. That is, with an increase in the environmental responsibility of the consumer, individuals are more likely to generate green consumption intentions and buy environmentally friendly clothing. Hence, we assumed the following hypothesis.

Hypothesis 6 (H6). Environmental responsibility has a significant effect on green consumption intention.

Considering the close relationship between environmental responsibility and green consumption intention, a chain mediating effect may exist between the environmental values and green consumption of apparel, where environmental values can indirectly influence green consumption intention via environmental responsibility, thus influencing the green consumption of apparel. To the best of our knowledge, the chain of environmental values–environmental responsibility–green consumption intention–green consumption of apparel has not been explored in previous studies.

In the hypotheses stated above for this study, we have argued the role of environmental responsibility in mediating between environmental values and green consumption of apparel, i.e., the transmission path of "environmental values–environmental responsibility–green consumption of clothing". Therefore, can green consumption intention further mediate the influence of the path from environmental responsibility to green apparel consumption? It has been demonstrated that there is a mediating path (environmental concern) between environmental responsibility and green consumption behavior [77]. Wang et al. [78] revealed the significant and positive mediating effect of behavioral intention on environmental responsibility and sustainable consumption behaviors when studying the sustainable consumption behaviors of rural residents in China. Therefore, we inferred that the transmission path exists, i.e., "environmental responsibility-green consumption intention-green consumption of clothing". Based on the above analysis, the following hypotheses were proposed.

Hypothesis 7 (H7). *Environmental responsibility and green consumption intention play a chain mediating role between environmental values and the green consumption behavior of apparel.*

Hypothesis 7a (H7a). Environmental responsibility and green consumption intention play a chain mediating role between egoistic values and the green consumption behavior of apparel.

Hypothesis 7b (H7b). Environmental responsibility and green consumption intention play a chain mediating role between altruistic values and the green consumption behavior of apparel.

Hypothesis 7c (H7c). *Environmental responsibility and green consumption intention play a chain mediating role between ecological values and the green consumption behavior of apparel.*

Based on the above discussion, the green consumption behavior of the apparel chain mediating model was constructed as shown in Figure 1.



Figure 1. The proposed theoretical model (Note: The unlabeled H3, H5, and H7 in this figure represent the mediating effect and chain mediating effect of environmental responsibility and green consumption intention).

The relevant models of green apparel consumption behaviors related to environmental values, environmental responsibility, and green consumption intentions are summarized in Table 1. Obviously, there is no systematic and in-depth study on the transmission paths between environmental values and green apparel consumption behavior, especially the chain mediation path containing environmental responsibility and green consumption intention. The theoretical model proposed in this study contributes to a fulfillment of the research gap in the literature on environmental values and green consumption behavior of apparel.

Authors	Independent Variable	Dependent Variable	Mediating Variable	Moderate Variable	Research Subjects
[1]	Value (fashion involvement)	Willingness to pay more for eco-apparel	Environmental knowledge, environmental concern, environmental behaviors	-	Undergraduate students in the USA
[25]	Past environmental behavior, green peer influence, green apparel knowledge	Green apparel purchase behavior	Perceived benefit of green apparel	-	Indian consumers
[33]	Environmental value, epistemic value, functional value, conditional value, social value, emotional value	Consumer behavior regarding green products	-	-	Polish consumers

Table 1. Summary of previous major studies and their models.

Table 1. Cont.

Authors	Independent Variable	Dependent Variable	Mediating Variable	Moderate Variable	Research Subjects
[36]	Environmental knowledge, general environmental attitudes, perceived money availability, perceived store accessibility	Actual buying behavior of sustainable apparel	Apparel environmental attitudes, subjective norms, perceived behavioral control, behavioral intentions	-	College students in USA
[72]	Knowledge, environmental concern, attitudes of environmentally sustainable apparel CSII(Consumer susceptibility	Purchasing behavior of environmentally sustainable apparel	-	-	Kuwaiti women
[79]	to interpersonal influence)-normative, CSII-informative, environmental apparel knowledge	Green apparel consumption behavior	General environmentally responsible behavior	-	Students (age 18-30) of Midwestern University, United States
[52]	Environmental value	Purchase satisfaction, repurchase intention of Eco-friendly fashion	-	-	Consumers in Korea
[56]	Environmental value	products Environmental responsibility	- Environmontolicm	-	Young people in Serbia
[80]	Environmental apparel knowledge	Environmentally responsible apparel consumption behavior	(internal, substantive, external), materialism (success, central- ity/acquisition, happiness)	-	Students at Midwestern University, United States
[22]	Convenience, information availability, selection, customized offerings, trend, social, adventure, authority, and status	Green apparel purchase intention	Shopping motivation (utilitarian and hedonic motivation)	Sex, family income	India consumers aged from 15 to 65 years old
[43]	Green communication, altruism, and openness to change	Purchase intention towards green apparel	Attitude	-	Young consumers in India
[61]	Compensatory health beliefs, environmental self-identity, and perceived environmental responsibility	Pro-environmental behavior	Perceived environmental responsibility	-	Nutricosmetics consumers in Malaysia
[26]	Green consumption values and social influence	Purchase intention for organic clothing	Attitude	-	Young consumers in India
[73]	Chinese cultural values (specifically, the Doctrine of the Mean)	Green purchasing intention	Lifestyles of consumers (fashion consciousness, leadership, cost consciousness, and development consciousness)	Environmer knowl- edge	ntal Chinese consumers
[77]	Environmental responsibility	Green consumption intention	Environmental concern	Price sen- sitivity	Chinese consumers

Authors	Independent Variable	Dependent Variable	Mediating Variable	Moderate Variable	Research Subjects
[78]	Environmental value, environmental knowledge, environmental responsibility, environmental sensitivity, response efficacy, perceived behavioral control, perception of consequence	Sustainable consumption behavior	Behavioral intention	-	Rural residents in China
[70]	Brand experience, attitude towards green purchasing, perceived behavioral control, social influence, supportive behavior towards	Purchase intention toward green apparel products	Customer engagement	-	Young consumers in India
[81]	environmental organization Ecological consumer awareness (perceived consumer effectiveness, environmental concern, clothing environmental attitude), product consumption value (emotional, social, epistemic, price, functional)	Bamboo product purchase intention	-	-	College students in USA

Table 1. Cont.

3. Research Methodology

To address the "perception–action paradox" of green apparel consumption among the Chinese Gen Z, this study aimed to construct a chain multiple mediation research framework and to explore the transmission path between environmental values and green apparel consumption behavior, as well as the associated influence mechanisms. This was performed by integrating environmental responsibility and green consumption intention into the analysis.

3.1. Research Design

Based on the above research objectives, the research framework comprised four procedures (see Figure 2). First, a comprehensive literature review was conducted to address the research questions, followed by the development of hypotheses, a research model, and taking of measurements. Second, a multistage sampling technique was used to ensure that the questionnaire was targeted to the nationwide population of Gen Z, who were born between 1995 and 2002 and had engaged in green clothing consumption behaviors. Third, data processing and analysis were conducted, including data cleaning, descriptive statistical analysis, common method bias test, reliability and validity analyses, correlation analysis, and chain mediation effect test. The conclusions and management insights will be presented.



Figure 2. The research roadmap.

3.2. Questionnaire and Instrument Development

The theoretical model proposed in this study contains six instruments in total, all of which are latent variables that cannot be directly measured. Therefore, a structured self-administered questionnaire with multiple items was developed using each instrument. The questionnaire comprises two main sections. The first section measures the respondents' demographic information, such as sex, education, monthly consumption, etc. The second section demonstrates the consumers' environmental values and green consumption behavior of apparel, as well as their environmental responsibility and green consumption intention.

To measure Gen Z's environmental value, we adopted the measurement scale developed by De Groot et al. [50] The scale comprises three dimensions, including the egoistic, altruistic, and biospheric values, in a total of 12 items measuring the consumers' overall environmental value. Environmental responsibility was assessed using 3 items adopted from Stone et al. [55]; green consumption intention was measured by 3 items adopted from Ajzen [13]. Three items evaluated the green consumption behavior of apparel, which was developed on our own based on the research results found in [32,36]. All items (see Table 2) were anchored on a five-point Likert scale, ranging from 1 (strongly disagree) to 5 (strongly agree), with 3 indicating neither agree nor disagree.

Variable		Items	Resource
Egoistic Values (EV)	EV1 EV2 EV3 EV4	Social power: control over others, dominance Wealth: material possessions, money Authority: the right to lead or command Influential: having an impact on people and events	
Altruistic Values (AV)	AV1 AV2 AV3 AV4	Social justice: correcting injustice, care for the weak A world at peace: free of war and conflict Equality: equal opportunity for all Helpful: working for the welfare of others	- Stern et al. [48], De Groot et al. [50,51]
Biospheric Values (BV)	BV1 BV2 BV3 BV4	Protecting the environment: preserving nature Preventing pollution: protecting natural resources Respecting the earth: harmony with other species Unity with nature: fitting into nature	-
Environmental Responsibility (ER)	ER1 ER2 ER3	Everyone has an obligation to take responsibility for the environment. I would rather sacrifice my personal interests to protect the environment. I have an obligation to discourage un-environmentally friendly consumption behavior.	Stone et al. [55]
Green Consumption Intention (GCI)	GCI1 GCI2 GCI3	I would like to collect and pay more attention to information concerning green products. I would like to buy and use green products to protect the environment. I would like to introduce and recommend green products to my family members and friends.	Ajzen [13]
Green Consumption Behavior of Apparel (GCBA)	GCBA1 GCBA2 GCBA3	It is well worth investing more money in green clothing. I often buy green clothing products at an affordable price. I often give priority to green clothing over conventional products.	Kim et al. [32], Chang and Watchravesrinkan [36]

Table 2. The measurement scale and resource.

Meanwhile, previous studies have shown that demographic variables of respondents such as sex, age, education, and expenditure have impacts on their consumption behavior [33,82]. Thus, we introduced sex, education, and monthly expenditure as control variables in this study to avoid differences between clusters caused by these variables.

3.3. Sampling Technique and Data Collection

Regarding the reliability and validity of the survey, all items measured in the six constructs of the questionnaire were developed on scales that have been proven in previous relevant studies and have been revised by consulting experts in the apparel field. Respondents were also invited to review and test the items repeatedly until the items in the questionnaire were logical and rigorous.

The formal questionnaire was carried out from April to August 2021. All questionnaires were anonymous, and our collected data included no personally identifiable information. To ensure a smoothly conducted survey process, the target population in this study focused on Chinese Gen Z members aged 18 years old and above, i.e., those born between 1995 and 2002, who are among the leading consumers with established values and independent apparel consumption and decision-making capabilities. Given the impact of the COVID-19 pandemic in China during this period, there were many difficulties for the field surveys. We adopted the snowball sampling method through the app WeChat, the most popular mobile social networking platform in China [83] that was reported to be used by more than three billion active users daily by the end of 2019 [84]. On the one hand, with the help of the WeChat links shared by friends, it was easy to find qualified interviewees in a quick, precise, and cost-effective manner to achieve a higher response rate [85]. On the

13 of 26

other hand, breaking through the geographical restrictions was possible to ensure that the interviewees involved came from the entire region of the country.

Eventually, a total of 803 questionnaires were collected, of which 146 questionnaires were deleted due to incomplete or illogical answers to questions; we retained 657 valid questionnaires and used them for the subsequent data analysis. The effective recovery rate was 81.8%. The survey respondents covered the eastern, central, and western regions of China. The survey sample is shown in Table 3, which shows that the male-to-female ratio was 47.8:52.2. The education level was mainly college or bachelor's degree, and the monthly consumption was concentrated below CNY 3000. This is mainly because the respondents of this study focus on Gen *Z*, who are independent consumers and are basically at the stage of studying in colleges or are simply working; therefore, the survey respondents' group characteristics such as age, education level, and monthly consumption amount are relatively concentrated. Regardless, they were eligible for this survey and have a certain level of representativeness.

Variable	Category	Frequency	Percent
<u>C</u>	Male	314	47.8%
Sex	Female	343	52.2%
	Elementary school and below	3	0.5%
E la contrar	Senior high school	24	3.7%
Education	Junior college and Bachelor's degree	564	85.8%
	Masters and above	66	10.0%
	Less than CNY 1500	406	61.8%
Monthly	CNY 1501-3000	226	34.4%
Consumption	CNY 3001-6000	16	2.4%
	More than CNY 6000	9	1.4%

Table 3. Summary of the demographic characteristics of respondents (n = 657).

4. Empirical Analysis

4.1. Common Method Biases Test

The green apparel consumption model constructed in this study serves as a multivariate analysis model involving six variables, such as egoistic values, altruistic values, biospheric values, environmental responsibility, green consumption intention, and green apparel consumption behavior. The single questionnaire self-assessment approach that was used in this study is prone to problems of common method bias (CMB) [86]; therefore, the questionnaire was anonymously administered and with reduced semantic ambiguity. Additionally, Harman's single-factor test was used to test the common method's deviation before data analysis to ensure the study's rigor.

Utilizing the SPSS 22.0 statistical analysis software, the results of the Harman's singlefactor test showed that six factors abstracted in a total explained variance of 79.689%. The explained variance of the first factor accounted for 20.665% of the total variance, far less than 50% [87]; there is no single factor or common factor that plays a major role in explaining the variance. In addition, the results of the common method bias test for the typical multifactor model fit are shown in Table 4. The single-factor model fit was very poor (χ^2 /df = 23.883, CFI = 0.618, NFI = 0.609, NNFI = 0.576, and RMSEA = 0.187), which presented no serious common method bias in this study.

Meanwhile, the fit results of the six-factor model were significantly better than that of the others, indicating that the six-factor model was superior, with good data matching and a better discriminant validity for the measurement of each variable scale (see Table 4). In the six-factor model, $\chi^2/df = 3.916 < 5$, CFI = 0.955, NFI = 0.941, and NNFI = 0.946, all of which were greater than 0.9; the model also had RMSEA = 0.067 < 0.08, indicating that the model constructed in this study has a good fit and is acceptable.

No.	Models	x ²	df	χ^2/df	RMSEA	CFI	NFI	NNFI
1	Single-factor model (EV + AV + BV + ER + GCI + GCBA)	4513.941	189	23.883	0.187	0.618	0.609	0.576
2	Three-factor model (EV + AV + BV, ER + GCI, GCBA)	2789.464	186	14.997	0.146	0.77	0.758	0.740
3	Three-factor model (EV + AV + BV, ER, GCI + GCBA)	2419.065	186	13.006	0.135	0.803	0.790	0.777
4	Four-factor model (EV + AV + BV, ER, GCI, GCBA)	2040.544	183	11.151	0.124	0.836	0.823	0.812
5	Five-factor model (EV, AV + BV, ER, GCI, GCBA)	1300.608	179	7.266	0.098	0.901	0.887	0.884
6	Five-factor model (EV, AV, BV, ER, GCI + GCBA)	1433.755	179	8.01	0.103	0.889	0.876	0.87
7	Six-factor model (EV, AV, BV, ER, GCI, GCBA)	681.436	174	3.916	0.067	0.955	0.941	0.946

Table 4. Results of common method bias test (n = 657).

Notes: df: degree of freedom; RMSEA: root mean square error of approximation; CFI: comparative fit index; NFI: normed fit index; NVFI: non-normed fit index; EV: egoistic values; AV: altruistic values; BV: biospheric values; ER: environmental responsibility; GCI: green consumption intention; GCBA: green consumption behavior of apparel.

4.2. Reliability and Validity Test

In this study, a Cronbach's alpha value was used to measure the reliability of each construct. The results of the reliability analysis (see Table 5) showed that the Cronbach's α values of egoistic values, altruistic values, biospheric values, environmental responsibility, green consumption intention, and green consumption behavior of clothing were 0.839, 0.867, 0.968, 0.826, 0.862, and 0.928, respectively, which were higher than the critical value of 0.7 [88], indicating that all scales passed the reliability test.

Tabl	le 5.	Relia	ability	' and	vali	dity	test.

Constructs	Items	Unstd.	S.E.	Z	<i>p</i> -Value	Std.	Cronbach's Alpha (α)	CR	AVE
	EV1	1	-	-	-	0.754			
Egoistic Values (EV)	EV2	0.831	0.050	16.697	0.000	0.674	0.020	0.046	0 591
	EV3	1.103	0.052	21.174	0.000	0.873	0.839	0.846	0.581
	EV4	0.965	0.053	18.330	0.000	0.736			
	AV1	1	-	-	-	0.849			
Altruistic Values (AV)	AV2	0.82	0.046	17.887	0.000	0.640	0.9/7	0.070	0 622
Altruistic values (AV)	AV3	1.019	0.037	27.399	0.000	0.863	0.867	0.872	0.633
	AV4	1.033	0.042	24.845	0.000	0.809			
	BV1	1	-	-	-	0.95	0.968	0.968	0.885
Biospharic Values (BV)	BV2	1.015	0.019	53.677	0.000	0.951			
biospheric values (bv)	BV3	1.022	0.021	48.801	0.000	0.932	0.968		
	BV4	0.958	0.02	48.283	0.000	0.929			
Environmontal	ER1	1	-	-	-	0.753			
Rosponsibility (FR)	ER2	1.248	0.063	19.879	0.000	0.841	0.826	0.831	0.621
Responsibility (ER)	ER3	1.148	0.062	18.614	0.000	0.768			
Croop Consumption	GCI1	1	-	-	-	0.719			
Intention (CCI)	GCI2	1.229	0.057	21.712	0.000	0.908	0.862	0.870	0.692
Intention (GCI)	GCI3	1.184	0.057	20.930	0.000	0.858			
Green Consumption	GCBA1	1	-	-	-	0.859			
Behavior of Apparel	GCBA2	1.059	0.032	33.023	0.000	0.925	0.928	0.929	0.815
(GCBA)	GCBA3	1.062	0.032	32.832	0.000	0.922			

Notes: CR: composite value; AVE: average variance extracted.

The constructs were tested for convergent and discriminant validity based on the reliability test. The CR values of all constructs (see Table 5) were higher than the critical

value of 0.7, and the AVE values of all constructs were greater than the critical value of 0.5. The square root of the AVE values of all constructs (see Table 6) was more significant than the Pearson correlation coefficients between each construct and other constructs in the conceptual model, indicating that each construct was statistically different from the others [88]. Thus, the results showed good reliability and convergent validity.

Table 6. Discriminatory validity test of the potential variables (n = 657).

Variables	Mean	S.E.	EV	AV	BV	ER	GCII	GCBA
Egoistic Values (EV)	3.879	0.665	0.762					
Altruistic Values (AV)	4.201	0.661	0.587 **	0.795				
Biospheric Values (BV)	4.628	0.664	0.451 **	0.703 **	0.941			
Environmental Responsibility (ER)	4.145	0.780	0.288 **	0.521 **	0.510 **	0.788		
Green Consumption Intention (GCI)	4.061	0.698	0.436 **	0.551 **	0.580 **	0.549 **	0.832	
Green Consumption Behavior of Apparel (GCBA)	3.839	0.788	0.303 **	0.463 **	0.462 **	0.485 **	0.551 **	0.903

Note: ** represents p < 0.01 (two-tailed). Diagonals (in bold) represent the square roots of the average variance extracted (AVE); the non-diagonal values represent the Pearson correlation coefficients.

4.3. Test of Correlation analysis

Results of the correlation analysis among the variables are also shown in Table 6, indicating a significant positive correlation between two of the constructs. Among them, egoistic values were significantly related to environmental responsibility (r = 0.288, p < 0.01), green consumption intention (r = 0.436, p < 0.001), and green consumption behavior of clothing (r = 0.303, p < 0.01); altruistic values had a significant relationship with environmental responsibility (r = 0.521, p < 0.01), green consumption intention (r = 0.551, p < 0.01), and green consumption behavior of clothing (r = 0.463, p < 0.01); biospheric values related to environmental responsibility (r = 0.510, p < 0.01), green consumption intention (r = 0.580, p < 0.01), and green consumption behavior of clothing (r = 0.462, p < 0.01). There was a significant positive relationship between environmental responsibility and green consumption intention (r = 0.549, p < 0.01) and green consumption behavior of clothing (r = 0.485, p < 0.01), and a significant positive relationship between green consumption intention and green consumption behavior of clothing (r = 0.551, p < 0.01). It is noteworthy that, compared with the altruistic and biospheric values, the correlation coefficients were lower between egoistic values and environmental responsibility, green consumption intention, and green consumption behavior of clothing.

4.4. The Direct Effects Test

The direct effects between the variables in this study's proposed model were verified using SPSS 22.0 software; the results of the multiple regression analysis of the model are shown in Table 7. Firstly, we tested the direct relationship between environmental values and green responsibility, green consumption intention, and green consumption behavior of apparel, i.e., whether H1, H2, and H4 were valid. As shown in Table 7, there was no significant direct effect of egoistic values on green consumption behavior of apparel ($\beta_{\rm EV} = 0.034$, p = 0.420). There was a significant positive effect of altruistic values and biospheric values on green consumption intention ($\beta_{AV} = 0.253$, $\beta_{BV} = 0.269$, p = 0.000), i.e., H1a is not valid, while H1b and H1c are confirmed; There was no significant effect of egoistic values on environmental responsibility ($\beta_{EV} = -0.043$, p = 0.286), while altruistic values and biospheric values had a significant positive effect on environmental responsibility (β_{AV} = 0.344, β_{BV} = 0.287, *p* = 0.000 < 0.001), i.e., H2a was not valid, while H2b and H2c are supported; All three of the egoistic values, altruistic values, and biospheric values had a significant positive effect on green consumption intention ($\beta_{EV} = 0.150$, $p = 0.000 < 0.001; \beta_{AV} = 0.203, p = 0.000 < 0.001; \beta_{BV} = 0.374, p = 0.000 < 0.001), i.e., H4a, p = 0.000 < 0.001$ H4b, and H4c were confirmed.

Variable	ER	GCI	GCBA	Results
Sex Education	0.014 - 0.014	0.019 -0.048	0.014 - 0.016	
Monthly Consumption Egoistic Values (EV)	$0.005 \\ -0.043$	0.024 0.150 ***	-0.020 0.034	H1a, H2a are not supported, while
Altruistic Values (AV) Biospheric Values (BV)	0.344 *** 0.287 ***	0.203 *** 0.374 ***	0.253 *** 0.269 ***	H1b, H1c, H2b, H2c, H4a, H4b,
F R ² Adjusted R ²	49.419 *** 0.313 0.307	70.617 *** 0.395 0.389	37.471 *** 0.257 0.250	supported.

Table 7. Regression analysis of variables (n = 657).

Note: *** represents p < 0.001; ER: environmental responsibility; GCI: green consumption intention; GCBA: green consumption behavior of apparel.

Secondly, we tested the direct effect between environmental values and green consumption intention (see Table 8), which showed that the environmental values significantly and positively affected green consumption intention ($\beta_{\text{ER}} = 0.489$, p = 0.000); thus, H6 is supported.

Table 8. Direct effect test of environmental responsibility (ER) on green consumption intention (GCI) (n = 657).

Variable	β	Т	<i>p</i> -Value	Results
Constant	2.121 ***	9.301	0.000	
Sex	0.040	0.879	0.380	
Education	-0.041	-0.706	0.480	
Monthly Consumption	0.016	-0.438	0.662	H6 is
Environmental Responsibility (ER)	0.489 ***	16.683	0.000	confirmed.
F		70.991 ***		
R ²		0.303		
Adjusted R ²		0.299		

Note: *** represents *p* < 0.001.

4.5. Chain Mediating Effect Test

In this study, two sequential mediating variables were included in the model, which formed three mediating paths: environmental values–environmental responsibility-green consumption behavior of apparel, environmental values–green consumption intention-green consumption behavior of apparel, and environmental values–environmental responsibility–green consumption intention–green consumption behavior of apparel. According to the test method and research results of the multistep mediating variables proposed by Hayes [89], the test of the mediating effect can be conducted regardless of the existence of main effects, and the key to the multistep mediating effect test is whether the mediating path with two sequential mediating variables is significant. Therefore, a bootstrap confidence interval test (5000 samples were selected at the 95% confidence level) was used in the SPSS 22.0 software to test the chain mediating effects of egoistic values, altruistic values, and biospheric values on green consumption behavior of apparel, respectively; each contained three paths, making a total of nine mediating paths. If the confidence interval of the mediating effect.

The chain mediation effect test results of egoistic values on green consumption behavior of apparel are shown in Table 9. It can be seen that the 95% confidence interval of egoistic values \rightarrow environmental responsibility \rightarrow clothing green consumption behavior was [0.0504, 0.1297], which did not include zero, indicating a significant indirect effect; The 95% confidence interval of egoistic values \rightarrow green consumption intention \rightarrow clothing green consumption behavior was [0.0928, 0.1864], illustrating a significant indirect effect; The 95%

confidence interval of egoistic values \rightarrow environmental responsibility \rightarrow green consumption intention \rightarrow clothing green consumption behavior was [0.0373, 0.0851], showing that the chain mediating effect was significant; thus, hypothesis H3a, H5a, and H7a are verified. Since the direct effect of egoistic values on green consumption behavior of apparel was not significant ($\beta = 0.0748$, [-0.0068, 0.1564] including zero); hence, environmental responsibility and green consumption intention completely mediate the relationship between egoistic values on green consumption behavior of apparel. The ratios of indirect effects to total effects for each mediating variable were 24.2% for the path of EV \rightarrow ER \rightarrow GCBA, 38.29% for the path of EV \rightarrow GCI \rightarrow GCBA, and 16.59% for the path of EV \rightarrow ER \rightarrow GCI \rightarrow GCBA, yielding a total indirect-to-total effect ratio of 79.08%. In comparison, the mediating effect of green consumption intention was the maximum, followed by environmental responsibility, and the minimum chain mediating effect of both variables.

Table 9. The chain mediation effect of EV on GCBA.

Paths	Effect	BootSE	BootLLCI	BootULCI	RTT/%	<i>p</i> -Value
Total effect	0.3575	0.0441	0.2710	0.4441	100	0.0000
Direct effect	0.0748	0.0415	-0.0068	0.1564	20.92	0.0722
Total indirect effect	0.2827	0.0337	0.2173	0.3503	79.08	
EV→ER→GCBA	0.0865	0.0202	0.0504	0.1297	24.20	
EV→GCI→GCBA	0.1369	0.0241	0.0928	0.1864	38.29	
EV→ER→GCI→GCBA	0.0593	0.0123	0.0373	0.0851	16.59	

Note: RTT presents the ratio to total effect.

The results of the chain mediation effect test of altruistic values on green consumption behavior of apparel (see Table 10). It can be seen that the 95% confidence intervals for altruistic values \rightarrow environmental responsibility \rightarrow green consumption behavior of apparel, altruistic values \rightarrow green consumption intention \rightarrow green consumption behavior of apparel, and altruistic values \rightarrow environmental responsibility \rightarrow green consumption intention \rightarrow green consumption behavior of apparel were [0.0698, 0.1954], [0.0996, 0.2062], and [0.0506, 0.1077], respectively; all of the intervals do not contain zero, indicating significant indirect effects. Thus, hypothesis H3b, H5b, and H7b are verified. The ratio of indirect effects to total effects for each mediating variable was 19.54% for the path of $AV \rightarrow ER \rightarrow GCBA$, 20.62% for the path of $AV \rightarrow GCI \rightarrow GCBA$, and 13.92% for the path of $AV \rightarrow ER \rightarrow GCI \rightarrow GCBA$, with a total indirect effect to total effect ratio of 64.87%. The mediating effect magnitudes, in descending order, were green consumption intention, environmental responsibility, and the chain mediation of both. The direct effect of altruistic values on green consumption behavior of apparel ($\beta = 0.1933$, p = 0.000) was significant and accounted for 35.13% of the total effect. Therefore, environmental responsibility and green consumption intentions partially mediate the relationship between altruistic values and green consumption behavior of apparel.

Paths	Effect	BootSE	BootLLCI	BootULCI	RTT/%	<i>p</i> -Value
Total effect	0.5497	0.0416	0.4680	0.6314	100.00	0.0000
Direct effect	0.1931	0.0471	0.1007	0.2855	35.13	0.0000
Total indirect effect	0.3566	0.0354	0.2879	0.4283	64.87	
AV→ER→GCBA	0.1300	0.0315	0.0698	0.1954	23.69	
AV→GCI→GCBA	0.1500	0.0274	0.0996	0.2062	27.12	
AV→ER→GCI→GCBA	0.0766	0.0143	0.0506	0.1077	13.92	

Note: RTT presents the ratio to total effect.

The chain mediation effect test results of biospheric values and green consumption behavior of apparel are shown in Table 11. It can be seen that the 95% confidence intervals for biospheric values \rightarrow environmental responsibility \rightarrow green consumption behavior of

apparel, biospheric values \rightarrow green consumption intention \rightarrow green consumption behavior of apparel, and biospheric values \rightarrow environmental responsibility \rightarrow green consumption intention \rightarrow green consumption behavior of apparel were [0.0748, 0.1980], [0.1168, 0.2214], and [0.0462, 0.0994], respectively; all intervals do not contain zero, indicating that they have significant indirect effects. Thus, hypothesis H3c, H5c, and H7c are verified. The ratio of indirect effects to total effects for each mediating variable was 24.15% for the path of BV \rightarrow ER \rightarrow GCBA, 30.21% for the path of BV \rightarrow GCI \rightarrow GCBA, and 12.82% for the path of BV \rightarrow ER \rightarrow GCI \rightarrow GCBA, with a total indirect effect to total effect ratio of 67.18%. The mediating effect magnitudes, in descending order, were green consumption intention, environmental responsibility, and the chain mediation of both. The direct effect of biospheric values on green consumption behavior of apparel ($\beta = 0.1803$, p = 0.0002) was significant and accounted for 32.82% of the total effect. Therefore, environmental responsibility and green consumption behavior of apparel.

Table 11. The chain mediation effect of BV on GCBA.

Paths	Effect	BootSE	BootLLCI	BootULCI	RTT/%	<i>p</i> -Value
Total effect	0.5494	0.0416	0.4676	0.6311	100.00	0.0000
Direct effect	0.1803	0.0416	0.4676	0.6311	32.82	0.0002
Total indirect effect	0.3691	0.0359	0.2989	0.4416	67.18	
BV→ER→GCBA	0.1327	0.0314	0.0748	0.1980	24.15	
BV→GCI→GCBA	0.1660	0.0268	0.1168	0.2214	30.21	
$BV \rightarrow ER \rightarrow GCI \rightarrow GCBA$	0.0704	0.0135	0.0462	0.0994	12.82	

Note: RTT presents the ratio to total effect.

In summary, we obtained a chain mediation model in which environmental responsibility and green consumption intention sequentially affect the relationship between environmental responsibility and green consumption behavior of apparel, and the relationship between the variables and the influence coefficients are shown in Figure 3.

Table 12 summarizes all the proposed hypotheses in this study, their implications for the theory of green apparel consumption and the practice of these findings will be discussed in the following section.

Нур	othesis	Hypothesized Path	β	<i>p</i> -Value	ab (95% BootCI)	Results
H1		ENV→GCBA	-	-	-	Partially supported
	H1a	EV→GCBA	0.034	0.420	-	Rejected
	H1b	AV→GCBA	0.253	0.000	-	Supported
	H1c	BV→GCBA	0.269	0.000	-	Supported
H2		ENV→ER	-	-	-	Partially supported
	H2a	$EV \rightarrow ER$	-0.043	0.286	-	Rejected
	H2b	$AV \rightarrow ER$	0.344	0.000	-	Supported
	H2c	$BV \rightarrow ER$	0.287	0.000	-	Supported
H3		$ENV \rightarrow ER \rightarrow GCBA$	-	-	-	Supported
	H3a	EV→ER→GCBA	-	-	0.0504-0.1297	Supported
	H3b	AV→ER→GCBA	-	-	0.0698-0.1954	Supported
	H3c	BV→ER→GCBA	-	-	0.0748-0.1980	Supported
H4		ENV→GCI	-	-	-	Supported
	H4a	EV→GCI	0.150	0.000	-	Supported
	H4b	AV→GCI	0.203	0.000	-	Supported
	H4c	BV→GCI	0.374	0.000	-	Supported

Table 12. Results of the proposed hypotheses tests.

Hypothesis Hyp	othesized Path	β	<i>p</i> -Value	ab (95% BootCI)	Results		
H5 ENV	−→GCI→GCBA	-	-	-	Supported		
H5a EV-	→GCI→GCBA	-	-	0.0928-0.1864	Supported		
H5b AV-	→GCI→GCBA	-	-	0.0996-0.2062	Supported		
H5c BV-	→GCI→GCBA	-	-	0.1168-0.2214	Supported		
H6	ER→GCI	0.489	0.000	-	Supported		
H7 $ENV \rightarrow$	ER→GCI→GCBA	-	-	-	Supported		
H7a EV→E	ER→GCI→GCBA	-	-	0.0373-0.0851	Supported		
H7b AV→E	ER→GCI→GCBA	-	-	0.0506-0.1077	Supported		
H7c $BV \rightarrow E$	ER→GCI→GCBA	-	-	0.0462-0.0994	Supported		
0.3354***	Environmental Responsibility 0.3186***	0.4114***	Green Consumptio Intention 0.2580***	n 0.4297***			
/							
Egoistic values				Green Consun	nption		
0	0.0	0748		Behavior of A	pparel		
0.6127***	Environmental Responsibility	(a) 0.3213***	Green Consumptio Intention	n 0.3890***			
		\searrow					
	0.3858***	\sim	0.2121***				
//							
Altruistic Values				Green Consun	nption		
	0.	.1931***		benavior of Aj	pparel		
(b)							
0.5987***	Environmental Responsibility	.3055***	Green Consumptio Intention	n 0.3847***			
	0.4316***	~	0.2216***				
Discultor 1 X7 1				Green Consur	nption		
Biospheric Values		1002***		Behavior of A	pparel		
	. 0.	1003		-			
		(\mathbf{c})					

Table 12. Cont.

Figure 3. The influence coefficients of the chain mediated model between environmental value and green consumption behavior of apparel. (a) Egoistic values, (b) Altruistic values, (c) Biospheric values. Note: *** represents p < 0.001.

5. Discussion

Over the past few years, greater apparel consumption and generation of waste have contributed to a growing negative impact on the environment. China urges all citizens to actively practice green consumption and as a result the people's environmental awareness has significantly grown. However, the "perception–action paradox" [34] still exists for the Chinese Gen Z, which presents a great challenge to China's sustainable development in the future. This study took Chinese Gen Z as its research subject, proposing a chain mediation model for the first time; this model aimed to find the relationship between environmental

values and green consumption behavior of apparel, as well as the transmission paths and their deep-seated influence mechanism. Based on the above study, the following findings were obtained.

Firstly, the results of this study confirm that the three types of environmental values, egoism, altruism, and biospheric values, have different direct effects on the green consumption behavior of clothing, which is in line with Bielawska and Grebosz-Krawczyk's research outcome [33]. However, we found that egoistic values do not have a direct significant influence on green consumption behavior of apparel for Gen Z in the Chinese context, which is inconsistent with the findings of previous scholars [50,51], namely, that environmental values significantly and negatively affect pro-environmental behavior. It implies that individuals with egoistic values are able to implement green consumption behaviors in members of Gen Z. This may be because Gen Z's environmental values are gradually changing with the increasing concern of society surrounding environmental issues. They have realized that their own lives are closely related to environmental protection, and they should develop their responsibility and green consumption intentions from their own interests. In other words, they do not perceive the conflict between implementing green consumption behavior of clothing and their own interests. On the contrary, they may gain "mianzi", social status, or their own influence in social groups because of green consumption behavior of clothing. In particular, when they perceive that the above benefits outweigh the investment cost of green clothing, they are very likely to implement green consumption behavior of clothing. This case is still grounded in the egoistic value orientation of individual self-interest, but it presents the possibility of actively engaging in the green consumption of clothing. Certainly, a large-scale empirical study in China is needed to further verify the above relationship.

Similarly, we conclude that both altruistic and biospheric values have a significant positive effect on the green consumption behavior of apparel, which further confirm the research findings of young consumers in India [43]. The direct effect of biospheric values on green consumption behavior of apparel is greater than that of altruistic values, which is consistent with previous research results [50,51]. This indicates that with the Chinese government's advocacy of sustainable consumption, Gen Z is more actively conscious of environmental issues than other generations. Whether individuals with altruistic values begin from others' interests or those with biospheric values start from ecological interests, they eventually actively participate in the green consumption of clothing and contribute to environmental protection as much as possible. In short, Gen Z members with the above three values will all be concerned about environmental issues and thus practice green consumption of clothing.

Secondly, the mediating effect of environmental responsibility and green consumption intention between environmental values and green consumption behavior of apparel was verified. The mediating effects of environmental responsibility and green consumption intention on the three types of environmental values are different, i.e., both play a fully mediating role between egoistic values and green consumption behavior of apparel, and a partially mediating role between altruistic values, biospheric values, and green consumption behavior of apparel. To the best of our knowledge, this finding is not available from previous studies. This means that there exists a transmission path of environmental values–environmental responsibility–green consumption behavior of apparel, and environmental values–green consumption intention–green consumption behavior of apparel in the relationship between environmental values and green consumption behavior of apparel. In contrast, the mediating effect of green consumption intention is greater than that of environmental responsibility, which suggests that we should pay more attention to green consumption intention; this finding is consistent with the results of previous studies which note intention as an essential mediating variable of behavior [13,74].

Finally, we reveal for the first time the chain mediating effect of environmental responsibility and green consumption intention between environmental values and green consumption behavior of clothing. Among these variables, the chain fully mediates the relationship between egoistic values and green consumption behavior of clothing, while partially mediating the relationship between altruistic values, biospheric values, and green consumption of clothing. This indicates that there is a transmission path of environmental values-environmental responsibility-green consumption intention-green consumption of apparel. Meanwhile, we confirmed that environmental responsibility significantly affects green consumption intention [77] and the significant and positive mediating effect of behavioral intention between environmental responsibility and sustainable consumption behaviors [78]. This means that when Chinese Gen Z is driven by environmental values, they are capable of thinking about the relationship between human and nature rights with respect to environmental issues, thus generating a stronger environmental responsibility and actively integrating it into their own behavior, allowing them to promote green consumption intentions and implement green clothing consumption behaviors. Therefore, the seamless connection between environmental values, environmental responsibility, green consumption intention, and green consumption behavior of clothing is extremely important. It can promote individuals' green clothing consumption habits and avoid the "inconsistency between words and actions". Consequently, the transformation efficiency of individuals' environmental values on green consumption behavior of clothing will be effectively enhanced.

6. Theoretical Contributions and Implications

6.1. Theoretical Contributions

This study first constructed a theoretical framework of environmental values and green apparel consumption behavior, exploring the determinants of green consumption of apparel and examining the influence mechanism of environmental responsibility and green consumption intention for Gen Z. The results of this study revealed the different direct effects of three environmental values (egoism, altruism, and biospheric value orientations) on the green consumption behavior of clothing of Gen Z in the Chinese context, which enriches the theoretical study of environmental values. In particular, the interpretation of egoistic values in the Chinese context further confirms the direct influence of cultural contextual factors on environmental values and the significant positive effects of altruistic and biospheric values on green consumption behavior. Meanwhile, the significant chain mediating effects of environmental responsibility and green consumption intention between environmental values and green apparel consumption behavior expand the transmission path and highlight new insights into the promotion of green consumption of apparel; the results provide theoretical support for the cultivation of environmental values in Chinese Gen Z as well as the sustainable development of Chinese society.

6.2. Managerial Implications

Based on the above discussion, the following insights are available for policy implications and sustainable consumption management in China, also providing some references for the promotion of sustainable consumption around the world.

First of all, it is indispensable to strengthen the publicity and education of environmental issues through the government-sanctioned environmental protection policies and education systems, and it is also important to cultivate good environmental values among Gen Z. Environmental values, similar to the standards and criteria of human perceptions and solutions to environmental problems, do not remain stable over the course of time, which gradually change with people's perceptions of the external environment. Values are more influential when activated in a particular situation, especially in the context of a severe environmental crisis. Therefore, the government should pay more attention to the dynamic changes of the environmental values of Gen Z and guide their behavior through comprehensive multilevel policies and through the education system, including school education, social education, corporate education, and family education. At the same time, we can make full use of the interactive and active characteristics of Gen Z on social media, specifying the importance of environmental values through the analysis of environmental pollution cases and motivating them to be more knowledgeable about sustainable living; this gives more priority to sustainable products and eventually forms biospheric values based on social or ecological interests. For Gen Z, which holds egoistic values, education should focus on the benefits brought to individuals by green clothing (such as the benefits of organic materials for the skin, social self-esteem coming from using green clothing, and social identity coming from green consumption). Thus, fashion designers and green apparel manufacturers should use eco-friendly fibers and fabrics to build consumer awareness of altruistic, physical, and aesthetic product attributes and link them to green self-identity, self-enhancement, and ecological awareness. Only when humans live in harmony with nature can individuals be qualified to pursue their own survival and living quality.

Secondly, it is imperative to enhance the environmental responsibility and perceived effectiveness of Gen Z by empowering and motivating the atmosphere through green consumption education. Currently, the main reason for the "perception-action paradox" of the green consumption of clothing is that consumers have insufficient knowledge about the green consumption of clothing and an insufficient perception of the effectiveness of their behavior. Therefore, the carbon label of apparel should be clearly marked to help consumers understand the negative impact of their own behavior on the environment and the environmental benefits they can bring by changing their own behavior, i.e., raising consumers' awareness of their responsibility by perceiving the harm and effectiveness of their own behavior, motivating them to connect their apparel consumption behavior with environmental issues, inspiring them to act in little ways, and changing their daily lifestyles by increasing their awareness of their environmental responsibility. Meanwhile, it is urgent to cultivate consumers' awareness of rational and reasonable consumption. We must encourage Gen Z individuals to share their green consumption experience and information on social media, build a positive green consumption community within their circle of friends, and promote the concept of green development. It is essential to change the aspect of Gen Z showing their social status and prestige through material ownership (e.g., luxury consumption) and instead have them be proud of green consumption, realizing their self-worth through sustainable lifestyles and actively practicing sustainable clothing consumption.

Finally, it is critical to strengthen the enterprise technological innovations of green clothing products and strengthen the low-carbon system reform; it is also important to build an institutional guarantee of a closed-loop system for green product consumption to increase the apparel green consumption intentions of Gen Z. The government should regulate the product development of apparel companies through green product systems and policies. According to the international green textile and apparel product certification, apparel enterprises are urged to increase the development of green technology for apparel and realize an eco-friendly method to the entire life cycle of apparel, from raw material production to apparel design, apparel manufacturing, apparel consumption, and disposal. At the same time, clothing enterprises are urged to establish green apparel development concepts that are centered on environmental value, led by green fashion, and are linked by emotional value, which satisfy Gen Z's needs for practicality, fashion, hedonism, and self-expression in apparel consumption. In addition, through technological innovation, it is necessary to reduce the price premium between green products and conventional products, which would inspire green consumption intention in the consumer and promote green consumption behavior in Gen Z consumers.

In brief, the greater the concerns for environmental issues, the stronger the moral obligation of consumers. The stronger the environmental responsibility of individuals, the more likely Gen Z consumers will activate their individual norms, thus enhancing their green consumption intentions and implementing altruistic and biospheric behaviors that correspond to their individual norms. That is, by strengthening the seamless linkage and transformation mechanism between environmental values, environmental responsibility, green consumption intention, and green consumption behavior of apparel among members of Gen Z, the green consumption of apparel will make significant progress in China.

7. Conclusions and Limitations

This study investigated the impact of Chinese Gen Z's environmental values on their green consumption of apparel. An important contribution of this research is its construction of a chain multiple mediation model between environmental values and green consumption of apparel that also integrates environmental responsibility and green consumption intention as mediators. We examined the direct and indirect effects of environmental values on green consumption of apparel in Chinese Gen Z members. The results revealed that all three types (egoism, altruism, and biospheric values) of environmental values have different direct and indirect effects on green apparel consumption behavior for Gen Z. In terms of the direct effects, egoistic values have no significant direct negative effect on green apparel consumption behavior, while the significant positive direct effect of biospheric values is greater than that of altruistic values. The total indirect effects of each (79.08%, 64.87%, and 67.18% of the overall effects, respectively) of the environmental values are much greater than their direct effects (20.92%, 35.13%, and 32.82% of the overall effects, respectively). Therefore, to address the "perception-action paradox" of Gen Z, it is important to concentrate on the mediating transmission path of environmental responsibility (24.20%, 23.69%, and 24.15% of the overall effects, respectively) and green consumption intention (38.29%, 27.12%, and 30.21% of the overall effects, respectively), as well as their chain mediating path (environmental values-environmental responsibility-green consumption intention-green clothing consumption behavior) (16.59%, 13.92%, and 12.82% of the overall effects, respectively); it is important to also concentrate on the mediating effect of green consumption intention. This is the first study to explicitly identify the significant chain mediating effect of environmental responsibility and green consumption intention between environmental values and green consumption behavior of clothing, which will broaden the theoretical perspective for the green consumption of apparel and guide green consumption practices and policy formulation for Gen Z worldwide.

Despite the contributions of this study, some limitations are worth mentioning for future research. Firstly, there are relatively more urban residents than rural residents in the survey sample, and our subsequent study expects to broaden the sample scope and size to enhance the representativeness of the sample. Secondly, social desirability bias may exist during the data collection, as green sustainability is widely advocated in China. Despite the anonymity used in the survey, individual respondents may be concerned that the expression of self-interested values may affect personal rights, social status, or influence, resulting in the over-reporting of socially desirable responses. Finally, this study only targeted Chinese Gen Z members, and it must be verified whether the proposed chain mediation path is applicable to consumers in other countries through international research. The main research directions in the future will aim to adopt diverse survey methods, increase the proportion of the survey sample for rural Gen Z, and conduct comparative studies of the urban–rural, intergenerational, and inter-country factors.

Author Contributions: J.L. (Jianfang Liang) developed the research idea, conceptualization, methodology, and wrote the manuscript; J.L. (Jingjun Li) wrote the draft manuscript, formal analysis, and performed the review and editing; Q.L. processed and curated some of the information and data. All authors have read and agreed to the published version of the manuscript.

Funding: This work was financially supported by The National Social Science Fund of China (Project No. 20XSH019).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The datasets generated during the current study are available from the corresponding author on reasonable request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Lee, S. Consumers' value, environmental consciousness, and willingness to pay more toward green-apparel products. *J. Glob. Fash. Mark.* **2011**, *2*, 161–169. [CrossRef]
- 2. The Nielsen Company. *The Sustainability Imperative: New Insights on Consumer Expectations;* The Nielsen Company: New York, NY, USA, 2018.
- 3. Daxue Consulting. Sustainable Consumption in China: Are Chinese Consumers Ready to Ride the Green Wave. Available online: https://daxueconsulting.com/sustainable-consumption-china/ (accessed on 9 September 2021).
- 4. Gleim, M.R.; Smith, J.S.; Andrews, D.; Cronin Jr, J.J. Against the green: A multi-method examination of the barriers to green consumption. *J. Retail.* **2013**, *89*, 44–61. [CrossRef]
- 5. Kumar, S.; Sadarangani, P. An empirical study on shopping motivation among generation Y Indian. J. Glob. Bus. Rev. 2021, 22, 500–516. [CrossRef]
- 6. Niinimäki, K.; Peters, G.; Dahlbo, H.; Perry, P.; Rissanen, T.; Gwilt, A. The environmental price of fast fashion. *Nat. Rev. Earth Environ. Behav.* **2020**, *1*, 189–200. [CrossRef]
- Cachon, G.P.; Swinney, R. The value of fast fashion: Quick response, enhanced design, and strategic consumer behavior. *Manag. Sci.* 2011, *57*, 778–795. [CrossRef]
- Ahmad, W.; Kim, W.G.; Anwer, Z.; Zhuang, W. Schwartz personal values, theory of planned behavior and environmental consciousness: How tourists' visiting intentions towards eco-friendly destinations are shaped? *J. Bus. Res.* 2020, 110, 228–236. [CrossRef]
- 9. Csanák, E. In Eco-friendly Concepts and Ethical Movements in the Fashion Industry. In Proceedings of the International Textile, Clothing and Design Conference—Magic World of Textiles, Dubrownik, Croatia, 5–8 October 2014.
- Hill, J.; Lee, H.H. Young Generation Y consumers' perceptions of sustainability in the apparel industry. J. Fash. Mark. Manag. 2012, 16, 477–491. [CrossRef]
- 11. Celik, G. Sustainable Fashion Is China's Fashion Industry Going Green. Available online: https://equalocean.com/analysis/2020 021613591 (accessed on 13 August 2020).
- 12. Statista. Global Apparel Market Size Projections 2012–2025. Available online: https://www.statista.com/statistics/279757 /apparel-market-size-projections-by-region/ (accessed on 23 July 2019).
- 13. Ajzen, I. The theory of planned behavior. Organ. Behav. Hum. Decis. Process. 1991, 50, 179–211. [CrossRef]
- 14. Bedard, S.A.N.; Tolmie, C.R. Millennials' green consumption behaviour: Exploring the role of social media. *Corp. Soc. Responsib. Environ. Manag.* **2018**, *25*, 1388–1396. [CrossRef]
- 15. Seemiller, C.; Grace, M. Generation Z: A Century in the Making; Routledge: London, UK, 2018.
- MacKenzie, J.; McGuire, R.; Hartwel, S. The First Generation of the Twenty-First Century. Available online: https://static1.squarespace.com/ static/56d7388222482e1e2c87c683/t/56e0cdc2cf80a14684670194/1457573327672/MagidPluralistGenerationWhitepaper.pdf (accessed on 20 May 2018).
- 17. Gentina, E.; Parry, E. *The New Generation Z in Asia: Dynamics, Differences, Digitalization;* Emerald Group Publishing: Bradford, UK, 2020.
- Chen, S. In Chinese Traditional Clothing Brands' Transformation in the Face of Generation Z—Taking Bosideng as an Example. In Proceedings of the 2022 2nd International Conference on Enterprise Management and Economic Development (ICEMED 2022), Dalian, China, 27–29 May 2022; Atlantis Press: Amsterdam, The Netherlands, 2022; pp. 35–39.
- 19. Sun, Y.; Wang, R.; Cao, D.; Lee, R. Who are social media influencers for luxury fashion consumption of the Chinese Gen Z? Categorisation and empirical examination. *J. Fash. Mark. Manag.* **2021**, *8*, 603–621. [CrossRef]
- 20. McCoy, L.; Wang, Y.-T.; Chi, T. Why Is Collaborative Apparel Consumption Gaining Popularity? An Empirical Study of US Gen Z Consumers. *Sustainability* **2021**, *13*, 8360. [CrossRef]
- Al Mamun, A.; Mohamad, M.R.; Yaacob, M.R.B.; Mohiuddin, M. Intention and behavior towards green consumption among low-income households. J. Environ. Manag. 2018, 227, 73–86. [CrossRef] [PubMed]
- 22. Kumar, S.; Yadav, R. The impact of shopping motivation on sustainable consumption: A study in the context of green apparel. *J. Clean. Prod.* **2021**, 295, 126239. [CrossRef]
- 23. Xu, Y. The influence of public self-consciousness and materialism on young consumers' compulsive buying. *Young Consum.* **2008**, *9*, 37–48. [CrossRef]
- 24. Bonini, S.; Oppenheim, J. Cultivating the green consumer. Stanf. Soc. Innov. Rev. 2008, 6, 56–61.
- 25. Khare, A. Green apparel buying: Role of past behavior, knowledge and peer influence in the assessment of green apparel perceived benefits. *J. Int. Consum. Mark.* **2019**, 1–17. [CrossRef]
- 26. Varshneya, G.; Pandey, S.K.; Das, G. Impact of social influence and green consumption values on purchase intention of organic clothing: A study on collectivist developing economy. *Glob. Bus. Rev.* 2017, *18*, 478–492. [CrossRef]
- 27. Yang, S.; Li, L.; Zhang, J. Understanding consumers' sustainable consumption intention at China's double-11 online shopping festival: An extended theory of planned behavior model. *Sustainability* **2018**, *10*, 1801. [CrossRef]
- Farbotko, C.; Head, L. Gifts, sustainable consumption and giving up green anxieties at Christmas. *Geoforum* 2013, 50, 88–96.
 [CrossRef]
- 29. Corraliza, J.A.; Berenguer, J. Environmental values, beliefs, and actions: A situational approach. *Environ. Behav.* **2000**, *32*, 832–848. [CrossRef]

- Lo, H.-W.; Liou, J.J.; Wang, H.-S.; Tsai, Y.-S. An integrated model for solving problems in green supplier selection and order allocation. J. Clean. Prod. 2018, 190, 339–352. [CrossRef]
- 31. Connolly, J.; Prothero, A. Green consumption: Life-politics, risk and contradictions. J. Consum. Cult. 2008, 8, 117–145. [CrossRef]
- 32. Kim, H.-S.; Damhorst, M.L. Environmental concern and apparel consumption. *Cloth. Text. Res. J.* 1998, 16, 126–133. [CrossRef]
- Bielawska, K.; Grebosz-Krawczyk, M. Consumers' choice behaviour toward green clothing. *Eur. Res. Stud. J.* 2021, 24, 238–256. [CrossRef]
- Liang, J.; Wang, R.; Li, J.J.S. Exploring the Relationship between Chinese Urban Residents' Perceptions of Sustainable Consumption and Their Efficiency Behavior: A Mediation and Moderation Analysis Based on the Social Practice Approach. Sustainability 2022, 14, 11262. [CrossRef]
- 35. Li, Y.; Lu, Y.; Zhang, X.; Liu, L.; Wang, M.; Jiang, X. Propensity of green consumption behaviors in representative cities in China. *J. Clean. Prod.* **2016**, *133*, 1328–1336. [CrossRef]
- 36. Chang, H.J.; Watchravesringkan, K.T. Who are sustainably minded apparel shoppers? An investigation to the influencing factors of sustainable apparel consumption. *Int. J. Retail Distrib. Manag.* **2018**, *46*, 148–162. [CrossRef]
- 37. Biswas, A.; Roy, M. Green products: An exploratory study on the consumer behaviour in emerging economies of the East. *J. Clean. Prod.* **2015**, *87*, 463–468. [CrossRef]
- 38. Chekima, B.; Wafa, S.A.W.S.K.; Igau, O.A.; Chekima, S.; Sondoh, S.L., Jr. Examining green consumerism motivational drivers: Does premium price and demographics matter to green purchasing? *J. Clean. Prod.* **2016**, *112*, 3436–3450. [CrossRef]
- Husted, B.W.; Russo, M.V.; Meza, C.E.B.; Tilleman, S. An exploratory study of environmental attitudes and the willingness to pay for environmental certification in Mexico. J. Bus. Res. 2014, 67, 891–899. [CrossRef]
- 40. Niinimäki, K. Eco-clothing, consumer identity and ideology. Sustain. Dev. 2010, 18, 150–162. [CrossRef]
- 41. Jung, S.; Jin, B. Sustainable development of slow fashion businesses: Customer value approach. *Sustainability* **2016**, *8*, 540. [CrossRef]
- 42. Khare, A. Antecedents to Indian consumers' perception of green apparel benefits. Res. J. Text. Appar. 2020, 24, 1–19. [CrossRef]
- Tewari, A.; Mathur, S.; Srivastava, S.; Gangwar, D. Examining the role of receptivity to green communication, altruism and openness to change on young consumers' intention to purchase green apparel: A multi-analytical approach. J. Retail. Consum. Serv. 2022, 66, 102938. [CrossRef]
- 44. Schwartz, S.H. Universals in the Content and Structure of Values: Theoretical Advances and Empirical Tests in 20 Countries. In *Advances in Experimental Social Psychology*; Zanna, M., Ed.; Academic Press: Orlando, FL, USA, 1992; Volume 25, pp. 1–65.
- 45. Schwartz, S.H. Are there universal aspects in the structure and contents of human values? J. Soc. Issues 1994, 50, 19–45. [CrossRef]
- 46. Dietz, T.; Fitzgerald, A.; Shwom, R. Environmental values. Annu. Rev. Environ. Resour. 2005, 30, 335. [CrossRef]
- Thøgersen, J.; Ölander, F. Human values and the emergence of a sustainable consumption pattern: A panel study. J. Econ. Psychol. 2002, 23, 605–630. [CrossRef]
- Stern, P.C.; Dietz, T.; Abel, T.; Guagnano, G.A.; Kalof, L. A value-belief-norm theory of support for social movements: The case of environmentalism. J. Hum. Ecol. Rev. 1999, 6, 81–97.
- 49. De Groot, J.I.; Steg, L. Value orientations and environmental beliefs in five countries: Validity of an instrument to measure egoistic, altruistic and biospheric value orientations. *J. Cross Cult. Psychol.* **2007**, *38*, 318–332. [CrossRef]
- 50. De Groot, J.I.; Steg, L. Value orientations to explain beliefs related to environmental significant behavior: How to measure egoistic, altruistic, and biospheric value orientations. *J. Environ. Behav.* **2008**, *40*, 330–354. [CrossRef]
- 51. De Groot, J.I.; Steg, L. Relationships between value orientations, self-determined motivational types and pro-environmental behavioural intentions. *J. Environ. Psychol.* **2010**, *30*, 368–378. [CrossRef]
- 52. Hong, B.-S.; Lee, E.-J.; Chu, Y.-J.J.F. The effect of environmental values of fashion consumers on purchase satisfaction and repurchase intention of eco-friendly fashion products. *Fash. Text. Res. J.* **2010**, *12*, 431–438. [CrossRef]
- 53. Holtbrügge, D.; Dögl, C. How international is corporate environmental responsibility? A literature review. J. Int. Manag. 2012, 18, 180–195. [CrossRef]
- 54. Wen, D.; Xiao, T.; Dastani, M. Pricing and collection rate decisions in a closed-loop supply chain considering consumers' environmental responsibility. J. Clean. Prod. 2020, 262, 121272. [CrossRef]
- Stone, G.; Barnes, J.H.; Montgomery, C. Ecoscale: A scale for the measurement of environmentally responsible consumers. *Psychol. Mark.* 1995, 12, 595–612. [CrossRef]
- 56. Slavoljub, J.; Zivkovic, L.; Sladjana, A.; Dragica, G.; Zorica, P.S. To the environmental responsibility among students through developing their environmental values. *Procedia-Soc. Behav. Sci.* 2015, 171, 317–322. [CrossRef]
- 57. Barr, S.; Gilg, A.W. A conceptual framework for understanding and analyzing attitudes towards environmental behaviour. *Geogr. Ann. Ser. B Hum. Geogr.* **2007**, *89*, 361–379. [CrossRef]
- 58. Hines, J.M.; Hungerford, H.R.; Tomera, A.N. Analysis and synthesis of research on responsible environmental behavior: A meta-analysis. J. Environ. Educ. 1987, 18, 1–8. [CrossRef]
- 59. Wu, B.; Yang, Z. The impact of moral identity on consumers' green consumption tendency: The role of perceived responsibility for environmental damage. *J. Environ. Psychol.* **2018**, *59*, 74–84. [CrossRef]
- 60. Kaiser, F.G.; Scheuthle, H. Two challenges to a moral extension of the theory of planned behavior: Moral norms and just world beliefs in conservationism. *Personal. Individ. Differ.* **2003**, *35*, 1033–1048. [CrossRef]

- 61. Shahrin, R.; Quoquab, F.; Mohammad, J.; Jamil, R. Factors affecting consumers' pro-environmental behaviour in nutricosmetics consumption: The role of perceived environmental responsibility as a mediator. *J. Asia Bus. Stud.* **2020**, *14*, 671–689. [CrossRef]
- 62. Rashid, N. Awareness of eco-label in Malaysia's green marketing initiative. Int. J. Bus. Manag. 2009, 4, 132–141. [CrossRef]
- 63. Jabeen, G.; Ahmad, M.; Zhang, Q. Perceived critical factors affecting consumers' intention to purchase renewable generation technologies: Rural-urban heterogeneity. *Energy* **2021**, *218*, 119494. [CrossRef]
- 64. Fishbein, M.A.; Icek, A. Predicting and Changing Behavior: The Reasoned Action Approach; Psychology Press: London, UK, 2011.
- 65. Yadav, R.; Pathak, G.S. Young consumers' intention towards buying green products in a developing nation: Extending the theory of planned behavior. *J. Clean. Prod.* **2016**, *135*, 732–739. [CrossRef]
- 66. Chi, N.T.K. Understanding the effects of eco-label, eco-brand, and social media on green consumption intention in ecotourism destinations. *J. Clean. Prod.* 2021, 321, 128995. [CrossRef]
- 67. Wang, L.; Wong, P.P.W.; Alagas, E.N. Antecedents of green purchase behaviour: An examination of altruism and environmental knowledge. *Int. J. Cult. Tour. Hosp. Res.* **2020**, *14*, 63–82. [CrossRef]
- Amoako, G.K.; Dzogbenuku, R.K.; Abubakari, A. Do green knowledge and attitude influence the youth's green purchasing? Theory of planned behavior. *Int. J. Product. Perform. Manag.* 2020, 69, 1609–1626. [CrossRef]
- Cheung, M.F.; To, W.M. An extended model of value-attitude-behavior to explain Chinese consumers' green purchase behavior. J. Retail. Consum. Serv. 2019, 50, 145–153. [CrossRef]
- 70. Joshi, Y.; Srivastava, A.P. Examining the effects of CE and BE on consumers' purchase intention toward green apparels. *Young Consum.* **2019**, *21*, 255–272. [CrossRef]
- Nguyen, T.N.; Lobo, A.; Greenland, S. Pro-environmental purchase behaviour: The role of consumers' biospheric values. *J. Retail. Consum. Serv.* 2016, 33, 98–108. [CrossRef]
- Kumar, B.; Manrai, A.K.; Manrai, L.A. Purchasing behaviour for environmentally sustainable products: A conceptual framework and empirical study. J. Retail. Consum. Serv. 2017, 34, 1–9. [CrossRef]
- 73. Sheng, G.; Xie, F.; Gong, S.; Pan, H. The role of cultural values in green purchasing intention: Empirical evidence from Chinese consumers. *Int. J. Consum. Stud.* **2019**, *43*, 315–326. [CrossRef]
- Albloushy, H.; Hiller Connell, K.Y. Purchasing environmentally sustainable apparel: The attitudes and intentions of female Kuwaiti consumers. *Int. J. Consum. Stud.* 2019, 43, 390–401. [CrossRef]
- 75. Sarkar, J.G.; Sarkar, A.; Yadav, R. Brand it green: Young consumers' brand attitudes and purchase intentions toward green brand advertising appeals. *Young Consum.* **2019**, *20*, 190–207. [CrossRef]
- Attaran, S.; Celik, B.G. Students' environmental responsibility and their willingness to pay for green buildings. *Int. J. Sustain. High. Educ.* 2015, 16, 327–340. [CrossRef]
- 77. Yue, B.; Sheng, G.; She, S.; Xu, J. Impact of consumer environmental responsibility on green consumption behavior in China: The role of environmental concern and price sensitivity. *Sustainability* **2020**, *12*, 2074. [CrossRef]
- Wang, P.; Liu, Q.; Qi, Y. Factors influencing sustainable consumption behaviors: A survey of the rural residents in China. J. Clean. Prod. 2014, 63, 152–165. [CrossRef]
- 79. Sadachar, A.; Khare, A.; Manchiraju, S. The role of consumer susceptibility to interpersonal influence in predicting green apparel consumption behavior of American youth. *Atl. Mark. J.* **2016**, *5*, 1.
- Sadachar, A.; Feng, F.; Karpova, E.E.; Manchiraju, S. Predicting environmentally responsible apparel consumption behavior of future apparel industry professionals: The role of environmental apparel knowledge, environmentalism and materialism. *J. Glob. Fash. Mark.* 2016, 7, 76–88. [CrossRef]
- 81. Yoo, J.-J.; Divita, L.; Kim, H.-Y. Environmental awareness on bamboo product purchase intentions: Do consumption values impact green consumption? *Int. J. Fash. Des. Technol. Educ.* 2013, *6*, 27–34. [CrossRef]
- Chen, L.; Zheng, H.; Shah, V. Consuming to Conserve: A Multilevel Investigation of Sustainable Consumption. *Sustainability* 2021, 14, 223. [CrossRef]
- Montag, C.; Becker, B.; Gan, C. The multipurpose application WeChat: A review on recent research. *Front. Psychol.* 2018, *9*, 2247. [CrossRef] [PubMed]
- Tencent. Tencent Announces 2019 Fourth Quarter and Annual Results. Available online: https://static.www.tencent.com/ uploads/2020/03/18/7fceaf3d1b264debc61342fc1a27dd18.pdf (accessed on 18 March 2020).
- Blair, J.; Czaja, R.F.; Blair, E.A. Designing Surveys: A Guide to Decisions and Procedures; Sage Publications: Thousand Oaks, CA, USA, 2013.
- Podsakoff, P.M.; MacKenzie, S.B.; Lee, J.-Y.; Podsakoff, N.P. Common method biases in behavioral research: A critical review of the literature and recommended remedies. J. Appl. Psychol. 2003, 88, 879. [CrossRef] [PubMed]
- Livingstone, L.P.; Nelson, D.L.; Barr, S.H. Person-environment fit and creativity: An examination of supply-value and demandability versions of fit. J. Manag. 1997, 23, 119–146. [CrossRef]
- 88. Hair, J.F.; Black, W.C.; Babin, B.J.; Anderson, R.E. Multivariate Data Analysis; Pearson Education: Hoboken, NJ, USA, 2010.
- Hayes, A.F. Introduction to Mediation, Moderation, and Conditional Process Analysis: A Regression-Based Approach; Guilford Publications: New York, NY, USA, 2013.